
Quickie status
Not much activity since snow has stopped tower and station work.

flux: Okay

tower: se sonics, TRHs, and DSMs fine.  50m nw sonic dead (need to check connector and possibly replace), 100m nw sonic is good, other nw sonics not 
mounted yet.

bao: Okay except EC150 which sends intermittently bad data.  Suspect that the terminal block screws inside EC100 box need to be tightened.  Other 
possibilities are the bulgin cable or the EC100 box itself, but the box was tested in the lab.

ehs: Have verbal permission for a site, but may need written permission.  Waiting for this decision (and snow) before deploying.  Kurt spent today 
preparing (which involves a few physical and cable changes to make a "cleaner" installation).

www: Created dat.Theta and dat.P on EOL servers.  List of www plots to generate is on Rudy's office marker board.  Julie is desperate for wind direction 
plots!
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